Author's response to reviews

Title: Fluorophotometry as a diagnostic tool for the evaluation of Dry Eye Disease

Authors:

Penny A Asbell (penny.asbell@mssm.edu)
Magid M Fahim (magid2000@hotmail.com)
Shamim A Haji (shamim.haji@mssm.edu)
Chak V Koonapareddy (chak.reddy@mssm.edu)

Version: 3 Date: 9 December 2005

Author’s response to reviews:

Abstract replaced with current formatted version according to instructions for authors. Minor essential revisions made according to reviewer's report - 1) fluorescein was changed to lower case on page 6 line 16, page 8 lines 8, 10 and 11.

2) "sever" was corrected to severe on page 9. 3) "may serves" changed to may serve on page 9 second line from bottom. 4) details of fluorescein corneal staining score added to manuscript. 5) Miyata et al. paper was added in the discussion section.